Author's response to reviews

Title: Infectious disease risk in asbestos abatement workers

Authors:

Lange H John (jhlange1@hotmail.com)
Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)
Luca Cegolon (l.cegolon@gmail.com)

Version: 5 Date: 18 July 2012

Author's response to reviews: see over
Dear Editor,

I am hereby re-submitting the following paper entitled: "Infectious disease risk in asbestos abatement workers", by Lange JH et al.

The structure of the manuscript and the references has been modified to comply with the journal’s formatting.

The comments of the reviewers have been addressed. In particular:

- For reviewer 1 (James Leigh): a yellow marked paragraph has been added in the Discussion section.

- For reviewer 2 (John Hoskins): two green marked sentences have been included to try to reword the concept;

Moreover few blue marked changes and few red marked erasures have been proposed (by the authors).

I look forward from you soon.

Best wishes.

Dr. Luca Cegolon, MD, MSc, FFPH
Corresponding author